Author's response to reviews

Title: Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund's adjuvant induced arthritic rat model

Authors:

Sanmugapriya Ekambaram (sanmug77@gmail.com)
Venkataraman Subramanian (svee1944@rediffmail.com)

Version: 8 Date: 6 August 2010

Author's response to reviews: see over
The article entitled “Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund’s adjuvant induced arthritis rat model MS no. 1343020502373125” is being resubmitted after carrying out the mentioned corrections

The following corrections have been carried out:

1. Authors list has been ordered as like in the submission system.

2. Authors contribution have been corrected as per the given instruction

3. The title and legend in the image page has been deleted and included in a separate page after references in the manuscript

4. The figures have been cropped as per the instructions and attached.

Thanking you

Thanks & regards

Dr.E.Sanmugapriya.